Author's response to reviews

Title: Nasopharyngeal cancer mimicking Otitic Barotrauma in a resource challenged centre: a case report.

Authors:

   Adekunle Daniel (kunle_d2002@yahoo.com)
   Ayotunde J Fasunla (ayofasunla@yahoo.com)

Version: 8 Date: 10 October 2011

Author's response to reviews:

Dear Editor In-Chief,

Dear Sir/Ma,


Thank you for your observations and comments. It is quite unfortunate that we cannot provide a clearer picture of figure 2. We feel that since we do not have a clearer picture, perhaps it is best to delete Figure 2 from the manuscript altogether.

This has been effected in the revised version sent forthwith.

Once again, we express our profound gratitude for your comments. We look forward to your response.

Yours faithfully,

Adekunle Daniel.